



Article EnergyAuction: IoT-Blockchain Architecture for Local Peer-to-Peer Energy Trading in a Microgrid

Felipe Condon ¹^(b), Patricia Franco ¹^(b), José M. Martínez ¹^(b), Ali M. Eltamaly ^{2,3}^(b), Young-Chon Kim ^{4,*}^(b) and Mohamed A. Ahmed ^{1,*}^(b)

- ¹ Department of Electronic Engineering, Universidad Técnica Federico Santa María, Valparaíso 2390123, Chile; felipe.condon.13@sansano.usm.cl (F.C.); patricia.franco@usm.cl (P.F.); jose.martinez@usm.cl (J.M.M.)
- ² Sustainable Energy Technologies Center, King Saud University, Riyadh 11421, Saudi Arabia; eltamaly@ksu.edu.sa
- ³ Electrical Engineering Department, Faculty of Engineering, Mansoura University, Mansoura 35516, Egypt
- ⁴ Department of Computer Engineering, Jeonbuk National University, Jeonju 561-756, Republic of Korea
- * Correspondence: yckim@jbnu.ac.kr (Y.-C.K.); mohamed.abdelhamid@usm.cl (M.A.A.)

Abstract: The widespread adoption of distributed energy resources (DERs) and the progress made in internet of things (IoT) and cloud computing technologies have enabled and facilitated the development of various smart grid applications and services. This study aims to develop and implement a peer-to-peer (P2P) energy trading platform that allows local energy trading between consumers and prosumers within a microgrid which combines IoT and blockchain technologies. The proposed platform comprises an IoT-cloud home energy management system (HEMS) responsible for gathering and storing energy consumption data and incorporates a blockchain framework that ensures secure and transparent energy trading. The proposed IoT-blockchain architecture utilizes a Chainlink oracle network and a private Ethereum blockchain. Through the use of smart contracts, consumers and prosumers can participate in an open auction to trade energy, while the settlement process involves acquiring external energy data from an API through the oracle network. The performance of the platform is evaluated through a testbed scenario using real-world energy data from a real house in Valparaiso, Chile, while storing those measurements in AWS cloud, validating the feasibility of the proposed architecture in enabling local energy trading. This work contributes to the development of energy management systems by providing a real-world implementation of an IoT-blockchain architecture for local energy trading. The integration of these technologies will allow for a more efficient and secure energy trading system that can benefit prosumers, consumers, and utilities.

Keywords: blockchain; local energy trading; microgrids; oracle networks; peer-to-peer

1. Introduction

Decades ago, traditional power grids were designed with the primary purpose of delivering electricity from large centralized power plants to individual residences. However, significant transformations have occurred in the electricity sector due to the integration of various technologies, including distributed energy resources (DERs), electric vehicles (EVs), advanced metering infrastructures (AMIs), and home energy management systems (HEMSs) [1]. DERs have enabled many end-users to transition from being energy consumers to becoming energy prosumers. This transformation allows prosumers to produce and consume energy simultaneously. In the near future, energy prosumers and consumers will be able to connect with one another and trade energy locally through a marketplace during instances of mismatch between supply and demand [2].

During the period when prosumers have surplus electricity, they can store it using energy storage systems (ESS), export it back to the power grid, or sell it to other customers. The direct energy trading among prosumers and consumers is called peer-to-peer (P2P) energy trading [3]. The implementation of this novel P2P energy trading structure allows



Citation: Condon, F.; Franco, P.; Martinez, J.M.; Eltamaly, A.M.; Kim, Y.-C.; Ahmed, M.A. EnergyAuction: IoT-Blockchain Architecture for Local Peer-to-Peer Energy Trading in a Microgrid. *Sustainability* **2023**, *15*, 13203. https://doi.org/10.3390/ su151713203

Academic Editor: Yang (Jack) Lu

Received: 10 July 2023 Revised: 23 August 2023 Accepted: 30 August 2023 Published: 2 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the local sharing of surplus energy from small-scale DERs within a microgrid, bringing advantages to prosumers, consumers, and the distribution system operator (DSO). P2P energy trading presents several benefits to all parties involved, as it facilitates local energy balance and improves the overall energy exchange process [4].

Internet of things (IoT) technologies are widely used to enable different smart grid applications. IoT technologies can be used for monitoring and controlling by using various sensors, actuators, and metering devices [5–8]. Such integrations can be found in HEMSs, which facilitates users' operation and management of household appliances and DER. To enable a microgrid application, such as trading within a microgrid, it is necessary to collect the energy consumption and/or generation data of houses. The use of these data serves as a key element to enable the implementation of various applications [9,10].

Blockchain technology has been gaining attention in non-cryptocurrency-related research, such as the integration of heterogeneous IoT systems with the energy market, with features such as data privacy, cybersecurity, and reliability [11]. In general, four layers compose the P2P energy trading system: the power grid layer, information and communication technology (ICT) layer, control layer, and business layer [12]. There have been several projects that have been carried out as energy trading pilots. Among these are the following projects: Piclo in the United Kingdom (see https://www.piclo.energy/ accessed on 2 November 2022), Transactive Grid and Brooklyn Microgrid in the USA [13], Peer Energy Cloud in Germany, and Powerledger in Australia [14,15]. The combination of IoT technology and blockchain technology has been covered in [16], where digitalization and decentralization are stated as the key enablers for a transactive energy internet (EI). A schematic diagram is presented in Figure 1 to provide a visual representation of the interaction between microgrids, distributed peer-to-peer networks, and cloud-based services. This diagram illustrates the main components of this research work, including the relationships and interactions between different entities. The microgrid is defined as a modern and innovative energy system that involves various parties, including electric vehicles (EVs), photovoltaic systems (PVs), energy storage systems (ESSs), prosumers, consumers, and utilities, which interact by exchanging both energy and data. In our previous work, Ref. [17], we presented the basic concept for the development of "EnergyAuction", a P2P energy trading platform that focuses on a proof of concept for the integration of oracles into smart contracts.



Figure 1. Schematic diagram of microgrid with distributed P2P networks and cloud-based services.

This work proposes the design and implementation of a P2P energy trading platform in a microgrid combining two key components to enable secure, transparent, and decentralized energy trading among participants: a HEMS and blockchain technologies. The HEMS facilitates real-time monitoring and control of energy consumption and generation and supports two different implementations based on edge devices and cloud-based services, respectively. The blockchain platform uses a private Ethereum network and Chainlink oracle network to facilitate energy trading between participants, storing energy transactions through a distributed ledger. Cloud-based services, such as the application programming interface (API), provide access to valuable energy consumption and generation data. To retrieve data from cloud-based services, oracle networks serve as a gateway, such as using available APIs, and route data to blockchain-based services.

Although a lot of research work has been conducted for peer-to-peer energy trading, there is no available information for the detailed technical implementation of such systems nor detailed aspects of such IoT–blockchain integration. The main contributions of this paper are given below:

- A thorough analysis of blockchain network design and market types/attributes for local peer-to-peer energy trading.
- Proposal of IoT–blockchain architecture for local P2P energy trading in a microgrid which consists of two layers: physical (IoT layer) and virtual (off-chain layer and blockchain layer).
- Development, deploying, and testing of four smart contracts for local peer-to-peer energy trading: data access contract, energy trading smart contract, Link Token contract, and oracle client contract.
- A testbed implementation (hardware/software) is carried out in a laboratory environment for one prosumer, one consumer, and a microgrid control center, where the off-chain layer of the HEMS is implemented on AWS platform and the blockchain layer is implemented using an AIO computer.

The remainder of this paper is organized as follows: Section 2 presents a review of the most relevant work related to blockchain technology, smart contracts, oracle networks, and the implementation of P2P energy trading. Section 3 describes necessary concepts for the design of P2P energy trading platforms, while Section 4 details the main components of the proposed architecture. Section 5 explains the process of setting up a market for P2P energy trading in a microgrid, including the definition of market type, archetypal market design, price formation mechanism, market participants, market value proposition, market scale, and market operation. Section 6 shows the implementation of the proposed platform, and in Section 7, the obtained results are analyzed. In addition, Section 8 provides the conclusions and future work.

2. Related Work

The energy internet presents a virtuous relationship among various entities, such as energy producers and end consumers. When these entities trade with one another, they are defined as a local energy market (LEM), which aims to support energy exchange between them in a competitive market and balance supply and demand locally [18]. The two key enablers for energy trading in a LEM are digitalization and decentralization [16]. Both decentralization and energy markets are presented below.

2.1. Decentralization: Blockchain Technology

The accelerated growth of renewable energy resources and the growing adoption of distributed power generation are transforming the energy sector. Peer-to-peer (P2P) energy trading has surfaced as an innovative solution, empowering prosumers and consumers to engage in energy trading in a more adaptable, efficient, and decentralized manner [19,20]. Blockchain technology, characterized by decentralization, transparency, and immutability, offers an ideal foundation for building P2P energy trading platforms [21].

Blockchain technology facilitates a secure and reliable environment for executing business logic and allowing participants in energy trading to interact. It creates transparent and tamper-proof records of energy transactions, ensuring that both producers and consumers can trust the system without relying on intermediaries [22]. This decentralized trust model, along with the ability to automate processes through smart contracts, has the potential to revolutionize energy trading and distribution [23]. Blockchain is also being introduced in the industrial internet of things (IIoT). For example, in [24], the authors used blockchain technology to enhance the security and privacy of a multi-camera multi-tracking using edge cameras. The authors in [25] presented an algorithm that improves the scalability and reduces latency. This improves the performance of blockchain for IIoT solutions. Numerous studies and projects have emphasized the potential of blockchain technology in facilitating P2P energy trading. For instance, the authors in [26] proposed a decentralized framework for coordinating energy supply and demand in a local market using blockchain and smart contracts. The work demonstrated that blockchain technology is a natural facilitator for implementing business logic and bringing energy trading participants together for interaction. As the energy sector continues to progress, the adoption of blockchain-based solutions is expected to rise, laying the foundation for a more decentralized and efficient energy landscape. General elements in the blockchain-enabled system consider network selection and configuration, consensus mechanism, privacy and network access, smart contracts, and oracle networks. These attributes are shown in Figure 2 and discussed in Section 2.2.



Figure 2. The blockchain system and its enabling elements.

2.2. Blockchain Network Design

2.2.1. Network Selection

Selecting the right blockchain network is essential for the success of an energy trading platform. There are several networks available, each with its unique features and characteristics. Ethereum (ETH), Hyperledger Fabric, Avalanche, and Cardano are some of the prominent options. The choice should be based on criteria such as the developer community, ecosystem maturity, and available tools. Ethereum is a popular choice due to its large developer base, robust ecosystem, and the backing of the Enterprise Ethereum Alliance. However, it is crucial to consider the specific requirements of the energy trading platform when selecting a blockchain network.

2.2.2. Consensus Mechanisms

A crucial aspect of the configuration of a blockchain network is choosing the appropriate consensus mechanism which ensures the reliability and trustworthiness of the system.

5 of 28

Popular consensus mechanisms include Proof of Stake (PoS), Proof of Work (PoW), Practical Byzantine Fault Tolerance (PBFT), and Delegated Proof of Stake (DPoS) [27]. The choice of consensus mechanism should be based on factors such as security, energy efficiency, and transaction throughput.

2.2.3. Privacy and Network Access

Several privacy and network access configurations can be considered based on the requirements of the energy trading platform. As an example, a public blockchain allows any participant to interact with the platform and join the network, whereas a permissioned or private blockchain restricts access to a selected group of users [27]. Privacy-preserving techniques, such as zero-knowledge proofs or secure multiparty computation, can be implemented to protect sensitive user data [28]. The selection of privacy and network access configuration should be aligned with the platform's objectives and the desired level of decentralization and security.

2.3. Smart Contracts

Smart contracts are a critical component of blockchain-based energy trading platforms, as they allow the automation of transactions and agreements between parties [29]. These self-executing contracts, coded with specific rules and conditions, eliminate the need for intermediaries and enhance the efficiency and transparency of the trading process. Designing and implementing robust smart contracts is essential to ensure the reliability, security, and performance of the platform. Several research papers and projects have demonstrated the use of smart contracts for energy trading. For example, the authors in [13] presented a study of a local energy market based on the Ethereum blockchain, where smart contracts were used to manage energy transactions between prosumers and consumers. The authors in [22] proposed a consortium blockchain-based energy trading framework that leverages smart contracts for trading and settling energy transactions between microgrids. This framework aimed to improve the efficiency and security of energy trading while reducing transaction costs. These examples highlighted the versatility of smart contracts in facilitating energy trading and their potential to transform the energy sector.

2.4. oracle Networks

Significant advancements have been made in the domain of blockchain-based networks. However, one notable limitation of smart contracts is their ability to interact with real-world data (also known as off-chain data). To overcome this challenge, the integration of blockchain oracles has become crucial. By incorporating blockchain oracles, the issue of securely and reliably incorporating external data sources into smart contracts can be effectively addressed. The authors in [30] conducted a survey aimed to identify researchers and institutions involved in blockchain oracle research. The survey findings revealed that a total of 162 papers were identified from different databases during the last seven years. This indicates that, despite the significance of the topic, there is still a need for greater efforts to establish a widely accepted oracle taxonomy. The results emphasize the importance of further research and collaboration in order to enhance the understanding and promote standardization in the field of blockchain oracles. In [31], different types of oracles were introduced to facilitate the selection process for specific blockchain-based applications. The authors in [32] introduced a decentralized energy flexibility market based on blockchain, which leverages oracles to fetch results from an off-chain computation process for determining flexibility matching. In this implementation, oracles were utilized to address the limited computational capabilities of smart contracts. In [33], the authors presented a decentralized access control system for IoT data that incorporates blockchain technology and trusted oracles. oracles were employed as intermediaries between entities responsible for hosting the blockchain network and IoT data.

2.5. Pilots and Implementations

Various pilots and testing platforms have been conducted to evaluate the potential of blockchain technology in energy trading. Table 1 summarizes the literature review of energy trading-related surveys, implementations, and testbeds. One notable example is the Brooklyn Microgrid project, which demonstrated the feasibility of a blockchain-based P2P energy trading platform within a microgrid [13]. Other examples include the Energy Web Foundation's EW-DOS platform, and the Power Ledger's trials in Australia [34]. These initiatives presented the potential benefits of blockchain technology in enhancing efficiency, transparency, and decentralization in energy trading while also highlighting the challenges that must be addressed for widespread adoption. In [21], the authors provided a comprehensive overview of blockchain's involvement in the energy sector, discussing its potential and limitations. The authors examined various scenarios where blockchain has been applied to energy trading, including mitigating the impact of electric vehicle charging during peak periods. The authors in [35], presented an energy trading implementation using IOTA, a "Tangle"-based framework, which is considered a third-generation DLT that uses a directed acyclic graph (DAG). The authors provided guidelines for deploying end-to-end transactions for the energy section on a dynamic set of scenarios.

This work differentiates itself from other related works by providing an implementation of a P2P energy trading platform using IoT and blockchain. The proposed implementation is validated in a testbed using real data consumption. Our work provides insight on how to implement and carry out a testbed scenario using the given architecture. Furthermore, a functional and cost analysis was also performed.

Table 1. Literature review for blockchain in smart grids. " \checkmark " represents "included", a cross mark " \times " represents "not included" in the reviewed articles.

Reference	Survey	Implementation	Testbed	Energy Trading
[13]	×	\checkmark	×	\checkmark
[21]	\checkmark	×	×	\checkmark
[34]	\checkmark	\checkmark	×	\checkmark
[28]	\checkmark	\checkmark	×	\checkmark
[35]	×	\checkmark	\checkmark	\checkmark
[36]	\checkmark	\checkmark	×	\checkmark
This work	×	\checkmark	\checkmark	\checkmark

3. Local P2P Energy Trading

In order to design and implement an energy trading platform, it is necessary to define the design of the energy trading market. The authors in [37] conducted a systematic literature review of three local energy market (LEM) models: P2P markets, collective self-consumption (CSC) or community, and transactive energy (TE) markets. The work provided detailed information on the market design and aspects of LEM transactions. These models were compared in terms of six categories representing attributes found in surveyed literature, archetypal market design, price formation mechanism, market participants, market value proposition, market scale, and market operation. The relationship between the three energy market models and the aforementioned attributes is shown in Figure 3.



Figure 3. Market types and attributes for energy trading.

3.1. Market Type

Three types of LEMs have emerged [20,37–39]:

- *P2P Markets*: Facilitate trading between users without the need for an intermediary. The primary objective of this market is to encourage active participation by energy users, providing them with an incentive to participate in the energy markets.
- *Collective Source Connected (CSC) or Collaborative*: Users tend to be co-located energy prosumers that trade their surplus energy in a market arrangement.
- *Transactive Energy Markets*: Maintain supply-demand equilibrium through autonomous management of DER using price signals, which ensures system stability. These markets offer services to the electricity system by optimizing the use of resources.

3.2. Archetypal Market Design

Six archetypal market designs have been identified [37,40–42]:

- *Futures market* is the most common market design, in which all the trading happens before the settlement period. When the settlement period arrives, the participants stick to their traded position. Any variation from the trading position is dealt with by the settlement.
- *Real time market* is where all the trading occurs at the settlement period, where participants contract with other participants to buy or sell the energy they are in the process of consuming or producing. This approach usually considers the usage of forecasted consumption or generation curves. One consideration found in this market design is that it is assumed that they are connected to larger traditional energy systems, which are able to provide or absorb an infinite excess supply or demand.
- *Mixed decentralized/centralized market* is a two-phase market design where initial decentralized trading is carried between participants looking to clear the market as far as possible, without intervention from a market operator. The second phase is performed by a centralized entity, which clears the remainder of the market.

- *Mixed futures/real time market* consists of two phases: An initial phase before the settlement, where participants trade based on predicted supply and demand for energy. During the settlement period, participants can adjust their position based on actual supply and demand curves in a real-time energy market.
- *Multi-layer market* starts from a local or bottom level for clearing the market. An aggregator is used for passing the uncleared supply or demand for energy to the next market level.
- *Settlement after the fact* is an uncommon scheme where no trading was carried out before the end of the settlement. Here, participants are charged or paid for the energy bought or sold after the settlement. A fixed price for buying or selling energy is defined by the system. Participants can purchase or sell the amount of energy they require.

3.3. Price Formation Mechanism

Price formation refers to the mechanism by which prices are discovered. There are five main categories [37,43]:

- *Single auction*: Only one party of the market is able to submit bids.
- *Double auction*: Allows sellers and buyers to submit bids.
- *System-determined mechanisms*: Usually defined by the system operator. Some of these mechanisms include uniform or fixed prices or cooperative game theory.
- *Negotiation-based mechanisms*: Automated transactions considered usually through the use of artificial intelligence (AI). Negotiating autonomous agents is one kind of negotiation-based mechanism.
- *Equilibrium-based mechanisms*: Use game theory to establish market equilibrium, which utilizes the bids and offers from the participants.

3.4. Market Participants

There are different market participants such as prosumers, consumers, and grid operators. Based on [2,37], the following are the main market participants:

- *Prosumers* are participants that generate and consume energy at the same time using small-scale distributed energy resources (DERs) such as wind turbines and solar panels.
- *Pure consumer* are participants that only consume energy without any DERs.
- *Pure generator* are participants that only generate energy.
- *Retailer* is an entity that connects to other large markets.
- *Market operator* is a single agent that runs the market operation.
- *Grid operator* is an entity that operates the electricity network and interacts with the market.
- *Aggregator* is an actor that represents a small group of participants.

3.5. Market Value Proposition

Market commodity and the benefits to participants are described in [37,44] as follows:

- *Market commodity* corresponds to the traded commodity in the market. Electrical energy was the most found commodity traded. Other commodities include flexibility alongside the sale of energy or a flexibility-only market. Other forms of energy are also found, such as combined heat and power, combined power and gas, or combined heat, power, and gas.
- *Benefits to market participants* are mainly financial, like increasing the profit from the sale of energy. Other objectives can also be considered, such as increasing the total welfare, which implies multiple financial benefits to all market participants, reducing the grid imbalance, or reducing the electricity cost.

3.6. Market Scale

Market scale considers the number of nodes involved in sizing the market [2,37]. For the participant scale, four different scales of participants are found:

- *Small scale*: Considers residential or individual energy users.
- *Building scale*: Multiple buildings trade with each other.
- Microgrid/community scale: Energy users operate as a community, rather than as individual participants.
- Grid scale: The trading is at a distribution or transmission network level.

3.7. Market Operation

Market operation refers to the type of data that is shared by the participants with the system, as well as the user preferences. Also, the settlement period and gate closures are discussed [20].

- Data shared: Price, volume, price and volume, demand and supply curve, controllable loads, the flexibility available, battery state of charge (SoC), distribution line distance, and discomfort level are some of the data parameters that can be shared for the energy trading.
- Settlement period and gate closure: The settlement period is a period of time in which the supply and demand of market participants are balanced. Gate closure is the period of time before the settlement, where bidding and offering can take place.

4. How Does Blockchain Enable Energy Trading?

Modern blockchain networks implement smart contracts which allow automatizing transactions [29,45]. Programming languages can be used to write scripts that implement energy trading business logic. This added to the benefits of an immutable, trustworthy, and decentralized network that provides the foundations to support an energy trading platform. As shown in Figure 4, a simplified diagram illustrates the interaction between the participants including prosumers, consumers, IoT devices, and the blockchain network. Data collected from the cloud-IoT HEMS is used to improve the implemented business logic in smart contracts. Information such as the available energy from an ESS, or the last hourly consumption from a householder, is used to enhance validations in the smart contract, resulting in viable trading scenarios. One challenge for using these data is the restriction for blockchain networks to fetch data from outside the blockchain. oracle networks work as a middleware between the blockchain networks and the data provider, as shown in Figure 5. oracle's primary job is to collect and provide data feeds to smart contracts [46].



Figure 4. Simplified diagram for interactions among prosumers, consumers, IoT devices, and the blockchain network.



Figure 5. IoT and blockchain interaction using oracles.

The proposed IoT–blockchain architecture for P2P energy trading in a microgrid is shown in Figure 6. The system is composed of multiple entities that are divided into two layers: the physical layer and the virtual layer. The main components of the system are the microgrid, cloud-based services, and distributed P2P networks. The microgrid consists of prosumers, consumers, battery energy storage systems, and photovoltaic systems, which exchange energy and data. Data related to energy consumption and generation are available through cloud-based services via APIs. Transactions of energy are recorded on a distributed ledger, such as a blockchain, through distributed P2P networks. oracle networks, another form of distributed P2P network, act as a connection between cloud-based services and blockchain-based services by retrieving data from APIs and routing them to the blockchain. The data that are routed back to the blockchain are then utilized by smart contracts.



Figure 6. IoT-blockchain architecture for P2P energy trading in a microgrid.

4.1. Physical Layer: IoT Layer

The physical layer considers the physical infrastructure that a microgrid requires to enable the energy trading feature. The microgrid is composed of three main actors: the prosumer, the consumer, and the microgrid control center (MCC). A prosumer is a consumer who has the capability of generating energy through photovoltaic systems or wind turbines. The energy stored in a battery energy storage system can be used as a power supply to satisfy its own consumption or to sell energy to the rest of the microgrid. Consumers consume the energy provided by the MCC, but they can also employ a BESS to store energy bought from the grid. The MCC manages the energy present in the microgrid. It also provides the power infrastructure for energy flow. This energy is bought or sold by households of the microgrid. Energy can be from decentralized energy resources or from the utility service provider. This will define the operation mode of the microgrid, by island mode or grid-connected mode, respectively.

The physical layer generates plenty of energy-related data, which are required to be stored and/or transmitted to the cloud layer in order to enable several smart grid applications. This challenge is addressed by IoT devices, which enable the collection of the intended energy measurements while also communicating the data to cloud-based services on the internet. The devices in the context of microgrids are smart meters. Smart meters are digital hardware devices that allow the reading of energy consumption from households on a regular basis. Other IoT-related devices are HEMSs, which aggregate energy-related data such as state of charge of BESS, appliance loads consumption, or the main household consumption.

4.2. Virtual Layer: Off-Chain Layer

The virtual layer of the IoT–blockchain architecture for P2P energy trading in a microgrid encompasses both energy data and energy trading transactions generated by the physical layer's IoT devices. This virtual layer is comprised of two sub-layers: the off-chain layer and the blockchain layer. The off-chain layer functions as a mediator between the real-world data captured by the IoT devices and the blockchain layer. The data are stored in cloud-based systems belonging to the HEMS provider and used for various purposes, such as load scheduling, load monitoring, alarm triggers, and forecasting. These features are available through cloud-based services offered by various HEMS providers. In addition, the MCC acts as a microgrid energy manager, collecting energy data from smart meters. This function enables the MCC to manage different processes and services on its own cloud servers. Both HEMSs and MCC cloud services may be self-hosted on-premise or through public cloud platforms such as Amazon Web Services (AWS), Google Cloud Platform (GCP), or Microsoft Azure. As these services manage energy storage data that are not stored on the blockchain, they are considered part of the off-chain layer.

4.3. Virtual Layer: Blockchain Layer

The blockchain layer plays a critical role in enabling secure and transparent energy trading in the microgrid. It provides a decentralized and trusted platform for energy transactions. The use of distributed ledger technology (DLT) ensures that all energy trading transactions are recorded in a tamper-proof and transparent manner. Furthermore, it enables all participants to have access to the same information and reduces the risk of fraudulent or malicious activities. The use of consensus mechanisms, such PoA, PoW, or PoS, enables the platform to validate and confirm energy trading transactions in a decentralized manner. This ensures that all parties involved in the energy trading platform can trust the outcome of the transaction and reduces the risk of fraud or errors.

5. Market Setup

This section provides a description of the market setup process for P2P energy trading in the microgrid. The following are the definitions of main elements, including the P2P market type, market design, price mechanism, types of market participants, market value proposition, market scale, and market operation. In order to establish a functional and efficient market, these key elements are crucial for P2P energy trading.

- Market Type: It is set up as a peer-to-peer, meaning that the participants are able to
 exchange energy directly among each other without intermediaries. This allows for
 greater autonomy and more flexibility in energy trading.
- Market Design: It is selected as a future market design, which means that the participants trade for future delivery of energy. Such a design allows for greater predictability and flexibility in energy trading.
- Price Formation Mechanism: The market price is determined by the intersection of the energy offered by the prosumer and the energy demanded by the consumer. This mechanism ensures efficiency and fairness in the energy trading process.
- Market Participants: The participants in the market are defined as prosumers, pure consumers, and the microgrid control center (MCC). The MCC is responsible for the overall operation of the microgrid and acts as a combination of both the market operator and the grid operator. The decentralization of the system through the use of blockchain technology ensures more efficient management of the microgrid.
- Market Value Proposition: The commodity traded in the market is energy, and the main benefit of the model is to put prosumers at the forefront of energy production and supply to consumers. This allows for a more sustainable and decentralized energy system.
- **Market Scale:** The market is defined as small-scale, with participants confined to a single microgrid. This allows for more efficient and localized energy trading.
- Market Operation: The market operation involves two categories: data related, including state of charge (SoC) data from the battery energy storage system (BESS), energy consumption and generation data from IoT devices, and trade-related, including price and quantity. The settlement and gate closure procedures are defined in detail in a subsequent figure, promoting a transparent and fair energy trading system.

Settlement and gate closure policy is depicted in Figure 7. The energy trading window and the energy trading sequence diagram are defined as part of the market setup. The trading window is composed of four sequential phases which make up the trading process:

- 1. **Bidding and Offering Period**: This is a 10 min phase in which consumers and prosumers can submit bidding and offer requests, respectively.
- 2. **Matching Period**: This is a 2 min phase in which the bids and offers are processed to make a match between them.
- 3. **Energy Transfer Period**: This is a 60 min phase during which energy is transferred from prosumers to consumers.
- 4. **Settlement Period**: This is a 3 min phase in which the settlement of funds takes place. The settlement period verifies how much consumers should pay to prosumers based on the actual energy received.



Figure 7. Trading window.

The four phases are modeled in the trading smart contract as valid states that a trading window can take. The transition between the states is executed autonomously by the microgrid control center using the smart contract when the expiration time elapses. Regarding the market operation, Figure 8 presents the interaction in which participants

manage to bid and offer energy. Before any trading operation is executed by the households, every householder has to grant permission to the trading smart contract to request energy data from its data sources, which is depicted as the data access granting phase from the trading window.



Figure 8. Test case sequence diagram for energy trading in a microgrid.

- 1. The MCC starts a new energy trading window, which begins with the bidding and offering phase. This is broadcasted as an event in the blockchain network.
- 2. The prosumer submits an offer to sell excess energy. The energy trading smart contract validates the offer request by obtaining energy readings from the HEMS through an oracle network. If the prosumer is unable to provide the offered energy, the smart contract emits an event to notify the prosumer that their offer will not be considered, and this is depicted in "Case 1". This ensures fairness and transparency by preventing

false offers. If the prosumer has enough energy, the smart contract records the energy offer as valid, and this is depicted in "Case 2".

- 3. A consumer submits a bid to buy energy, which is recorded in the record of bids. After 10 min, the smart contract matches available energy offerings with the record of bids, providing a match between prosumers and consumers for energy transfer.
- 4. In the energy-transferring phase, which lasts for 60 min, the prosumer becomes the energy provider of the consumer who charges its ESS with the energy provided. A maximum of 10 kWh is set for this phase due to technological restrictions of a common residential system to transfer more than 10 kW over an hour to the battery.
- 5. In the settlement phase, the energy trading smart contract obtains energy readings from the consumer smart meter on the HEMS API through an oracle network. If the consumer received less energy than they requested, funds are returned to cover the energy mismatch. The payment is then transferred to the prosumer based on the revenue generated from the sale of energy, which is determined by market demand through the use of bid pricing. The mathematical representation of this process can be represented as R = E * B, where R represents the revenue, E represents the energy received by the consumer, and B represents the bid price.

6. Implementation

The testbed was carried out in the research laboratory B110 at Universidad Tecnica Federico Santa Maria in Valparaiso, Chile. In the laboratory, two Raspberry Pi 4 Model B units were used to represent the prosumer and consumer participants. An all-in-one (AIO) computer was used to set the blockchain-related elements and for the MCC operations. Additional services were implemented in Amazon Web Services (AWS). As seen in Figure 9, the implementation is divided into three main components:



Figure 9. Testbed diagram.

6.1. Off-Chain Layer

The off-chain layer of the home energy management system (HEMS) was implemented on the AWS platform. AWS IoT Core was utilized to provide an MQTT broker for messaging between the energy data publisher microservice available in both the consumer and prosumer HEMS. This broker allows energy consumption and generation data to be sent to the cloud, where it is stored and made available through an API. Energy data for each household can be requested using a unique identifier assigned to each smart meter.

The two HTTP get endpoints implemented are:

- /battery-soc/id: This endpoint returns an object with the readings from the last measurement of the state of charge (SoC) of a battery energy storage system (BESS). The readings are randomly generated each hour, and the id represents the identifier of the battery.
- /smartmeter/id: This endpoint returns the last hour energy consumption in kWh from a user's smart meter. The id represents the identifier of the smart meter.

The implementation of the off-chain layer allows the collection and storage of energy data from both the consumer and prosumer HEMS. The use of AWS and the energy data publisher microservice ensures the feasibility of the IoT–blockchain energy trading system in microgrids. The source code for this implementation is provided in a GitHub repository. (https://github.com/pipegreyback/EViG-Server accessed on 3 November 2022).

6.2. Blockchain Layer

The blockchain layer of the IoT–blockchain energy trading system was implemented in a laboratory using an all-in-one (AIO) computer, which featured an AMD Ryzen 7 5300U CPU, 16 GB RAM, and a 1 TB hard drive. The Hyperledger Besu, an open-source Ethereum client, was used to implement a private Ethereum network. To speed up the network setup, Hyperledger Besu Docker quickstart was used, which allowed for the deployment of four validator nodes and three client nodes. The validator nodes were configured to operate using IBFT 2.0 consensus [47], which was chosen due to its high throughput and finality compared to other consensus mechanisms. This is also the consensus mechanism set by default in Hyperledger Besu. This consensus mechanism uses a dynamic mechanism for validator selection and requires at least a 2/3 validator signature in order to confirm a block. In case no consensus is reached, further validation rounds are required until the consensus is reached. Additional tools, such as a block explorer, were also provided to simplify the development process. Finally, to develop smart contracts, Remix, a popular online Solidity compiler, and a debugger were used.

6.2.1. Smart Contracts

Four smart contracts were deployed to enable energy trading. Figure 10 shows a summary of each smart contract by category.



Figure 10. Deployed smart contracts by category.

 In the privacy contract category, the data access contract was implemented to manage an access control list containing entities authorized to request energy data from a user's HEMS. This permission is granted by the user. This smart contract helps ensure that the energy data are only accessible to authorized parties, which helps to protect the privacy and security of the energy data.

- In the trading contract category, the energy trade smart contract was developed, which contains the main business logic of the energy trading system. This smart contract implements a trading window consisting of four states: a bidding and offering state, a matching state, an energy-transferring state, and a settlement state. During the initial phase, the bidding and offering period is initialized when a new trading window is requested in the trading smart contract. Additionally, the trading window allows granting data access at any time, which is required to participate in a trading window. A class diagram is presented in Figure 11, where the solidity structures are presented as classes, that possess attributes and functions.
- In the oracle contracts category, two contracts were developed. The Link Token contract issues the ERC-20 Link Token, which is required to pay for data requests through the oracle network. The oracle client contract is used to request energy data using the HEMS APIs, and this request is handled by the oracle network, which processes the request and provides a callback with the results of the request. Both the Link Token contract and the oracle client contract can be found in the Chainlink official repository (https://github.com/smartcontractkit/chainlink/tree/develop/contracts/src accessed on 3 November 2022).



Figure 11. Class diagram of energy trading contract.

6.2.2. oracle Nodes

In order to enable smart contracts to utilize the available API, the deployment of a Chainlink oracle network using docker containers was necessary. Additionally, a Post-greSQL database was deployed using docker containers to support the Chainlink oracle network. Configuration between the Ethereum network and the Chainlink oracle node was established to facilitate interaction between smart contracts and the oracle within the Ethereum network. The primary parameters utilized for communication between the two networks include the RPC URL and the chainld. Figure 12 shows the interaction diagram for the data feed request between smart contracts and oracle nodes.



Figure 12. oracle smart contract deployment and invocation diagram.

In order to access the endpoints provided by the HEMS API, it is necessary to create and deploy a job within the oracle node. A job comprises a sequence of steps, called tasks, that fulfill the data feed requirement of a smart contract. In this specific case, a direct request job was created. This job involved requesting data from the HEMS API, parsing the response, encoding the data into a format compatible with Solidity, and ultimately submitting the transaction back to the requesting smart contract. Figure A1 shows the direct request job configured.

6.3. IoT Layer

For the IoT layer, the two Raspberry Pi units were used. Docker was employed to run containers of the services required to enable the energy data client and the energy trading client.

6.3.1. Energy Data Client

The implementation of the IoT layer was done using the AWS platform and the energy data publisher microservice was developed using Node.js and MQTT.js. This microservice sends energy data to the HEMS cloud service using MQTT, allowing the collection and storage of energy data from each household. The microservice was packaged into a docker container for easy deployment and running on any operating system that supports docker containers.

6.3.2. Energy Trading Client

Both the consumer and the prosumer were configured with an energy trading client, which allows for submitting bids and offers to the blockchain to participate in the energy trading within the microgrid. The energy trading client was developed using Node.js and the ethers.js library, which is an Ethereum client that allows the execution of operations within a blockchain network. The energy trading client was also packaged into a docker container for ease of deployment. The trading client subscribes to the four events that the smart contract emits to the blockchain network. The event subscription and execution of smart contract functions are managed by ether.js. Table 2 describes the operations that any participant in the microgrid can execute within the energy trading smart contract. The client source code is provided in the repository, and it considers three possible configurations: consumer, prosumer, and MCC.

Operation	Description
Constructor	Initialize the contract's state variables when the smart contract is deployed.
Phase Transition	Update the current phase of the trading process based on a predetermined timeline or other criteria.
Start Trading Window	Mark the beginning of the trading phase and set the initial trading parameters, such as the duration of the trading window and the maximum amount of energy that can be traded.
Submit Energy Offer	Allow sellers to submit offers to sell their energy at a certain price and quantity. These offers are stored in the contract's state variables until they are matched with a corresponding bid.
Submit Energy Bid	Allow buyers to submit bids to buy energy at a certain price and quantity. These bids are stored in the contract's state variables until they are matched with a corresponding offer.
Matching Phase	Match the submitted offers and bids based on their price and quantity, and assign the matched pairs to a transfer list for the transferring phase.
Transferring Phase	Transfer the energy between the matched parties based on the agreed-upon price and quantity. This phase ensures that the energy is delivered securely and efficiently.
Settlement Phase	Transfer the payment between the matched parties based on the agreed-upon price and quantity, and update the state variables to reflect the completed trade. This phase ensures that the payment is made accurately and securely.

Table 2. Available operations for the energy trading smart contract.

6.4. Use Case and Validation

The testbed consists of one prosumer, one consumer, and a microgrid control center. We assumed that a prosumer owns a PV system and BESS, which will offer excess energy, the consumer with BESS will bid to fulfill his energy consumption requirements, and the microgrid control center will provide the power infrastructure for all households in the microgrid. The smart contract for energy trading was deployed by the lab all-in-one (AIO) computer acting as the MCC. Both the prosumer and consumer grant data access permission to the energy trading smart contract for making energy requests to the HEMS API. Figure A2 shows the event log where the prosumer executes the grant data access function passing the energy trading smart contract address and the smart meter id as the parameters in the invocation.

The energy trading window starts with an energy price of 60,000 gwei per kWh, which was set by the microgrid control center (MCC). This price is equivalent to around CLP 102 per kWh, which is the rate provided by energy utility Chilquinta in Chile (https: //www.chilquinta.cl/informacion-de-interes/tarifas-vigentes accessed on 3 November 2022) for the cost of energy per kWh, and the cost of transmission per kWh, for the BT1A tariff. The trading window has now entered the offering and bidding phase, allowing consumers and prosumers to submit their offers to sell or bids to buy energy at the specified price. The energy starting window event is broadcasted to all participants in the trading system, enabling them to react to the event on the clients by event subscription and execute their own state machines. This enhances the overall efficiency and transparency of the trading process, allowing participants to respond quickly and accurately to changes in the trading window and the energy price. The log message shown in Figure A3 confirms the current phase of the trading window and the energy price broadcasted in the event. This information is essential for understanding the initial pricing conditions and the current state of the energy trading system.

Later, a prosumer submits an offer to sell 10 kWh. This offer request triggers the oracle smart contract for making a request to the HEMS API for checking the prosumer available

energy. The event of an energy offering being denied due to insufficient energy is illustrated in Figure A4. An event with an error message indicating the reason for the denial, and in this case, "insufficient energy" is provided. In the event that a participant has not granted the energy trading smart contract the required permission to access the energy data, an error is returned by the contract validations, as seen in Figure A5.

In the case of valid energy offers and energy bids, when the matching phase finishes, a result event is broadcasted to the matched participants, as seen in Figure A6. This event indicates that the energy trading smart contract has reached the matching state, where a consumer and prosumer have agreed on a trade for 10 units of energy at a price of 54,000 gwei per unit. The smart contract has successfully matched the consumer's energy demand with the prosumer's energy supply at the agreed price, and the settlement phase can proceed.

Regarding the received energy by the consumer, Figure A7 shows the log from the oracle request. This log was emitted by the energy trading smart contract during the energy-transferring phase, indicating that the consumer received 8 kWh of energy from the prosumer, as measured by the smart meter. It is important to note that the energy data are managed by the Chainlink oracle, and the energy trading smart contract interacts with the oracle contracts to retrieve the data. The discrepancy between the initially agreed 10 kWh, and the 8 kWh received by the consumer may be due to a delay in the initial transfer from the prosumer, or external factors such as network congestion or power outages. However, the flexibility provided by the smart contract allows for mismatches between agreed energy and received energy to be addressed during the settlement phase.

The settlement phase, which follows the energy-transferring phase, is responsible for determining the proportional payment to be made to the prosumer based on the amount of energy received by the consumer. Figure A8 shows an example of the proportional payment calculation for the scenario where the consumer received 8 kWh of energy, which was less than the initially agreed 10 kWh. The implementation of the energy trading smart contract provides a degree of flexibility by allowing for adjustments to be made during the settlement phase in case of mismatches between agreed and received energy.

6.5. Cost Analysis Results

Table 3 shows a cost comparison of different energy trading smart contract operations performed in our energy trading platform and in various speed-based configurations on the public Ethereum network. As our implementation uses a private Ethereum network, there were no transaction fees incurred during contract deployment or execution. Therefore, the gas values listed in the table represent the resources used to perform the operations. In comparison, public Ethereum networks charge transaction fees for contract execution, and gas prices can vary based on the user's preference for transaction speed. This is because the public Ethereum network tends to become congested, where validators prioritize transactions with a higher gas price to process and receive a reward. As a result, users may choose to pay a higher gas price to increase the chances of their transaction being processed sooner. The public Ethereum network offers suggested gas prices for different transaction speeds, namely slow, average, or fast. For the public Ethereum network, each transaction speed has a corresponding gas price, which on the date of this study were 20 gwei, 22 gwei, and 24 gwei. To simplify the comparison, the transaction costs in the table are presented in Chilean pesos (CLP) using the conversion rate from ETH to CLP on 10 April 2023.

Block time in the Ethereum network is dynamic and reflects the network congestion and performance. In the case of Hyperledger Besu, the block time is a critical parameter that determines the block frequency. This time has a direct impact on both throughput and latency. If the block time is very short (e.g., 1 s), blocks will be generated frequently, which results in low latency and high throughput. However, in the case of a block period being very long, the performance will be reduced because the transactions will need to wait a long time to be added to the next block [48].

Contract Operation	Gas Utilization	Cost in CLP in Our System	Cost in CLP in Slow Public Ethereum	Cost in CLP in Average Public Ethereum	Cost in CLP in Fast Public Ethereum
Constructor	2,619,703	\$0	\$81.010	\$89.112	\$97.213
Phase Transition	161,524	\$0	\$4.995	\$5.494	\$5.994
Start Trading Window	286,403	\$0	\$8.857	\$9.742	\$10.628
Submit Energy Offer	129,613	\$0	\$4.008	\$4.409	\$4.810
Submit Energy Bid	130,354	\$0	\$4.031	\$4.434	\$4.837
Matching Phase	153,400	\$0	\$4.744	\$5.218	\$5.692
Transferring Phase	134,179	\$0	\$4.149	\$4.564	\$4.979
Settlement Phase	144,247	\$0	\$4.461	\$4.907	\$5.353

Table 3. Cost comparison of different energy trading smart contract operations based on the network execution.

In our implementation of the energy trading smart contract, the created Ethereum network provided a 5 s validation block time. In comparison, the last average block time of the public Ethereum network is around 12 s (https://etherscan.io/chart/blocktime accessed on 10 January 2023), as shown in Figure 13. The difference in block times could be due to various factors, such as network congestion or the consensus mechanism used by the network. Understanding the impact of block time on the energy trading smart contract's performance, throughput, and gas usage is critical for future optimizations and improvements.





Figure 13. Block time comparison between EnergyAuction network and public Ethereum.

The maximum block size in gas units for a regular block in our implementation and in public Ethereum networks is 30 million gas units. Figure 14 shows a bar chart where the eight operations of the energy trading smart contract are presented in terms of the percentage of block usage. The constructor is the most gas-consuming operation and

Network Congestion Level

utilizes 8.73% of the available gas in a block. The remaining operations present lower gas consumption, between 0.43% and 0.95% of gas usage for the available gas in a block. An analysis of the gas usage for the energy trading smart contract revealed that no more than 12 constructors or 200 trading operations could be performed within a single block. This indicates that the platform may face limitations in terms of scalability and network congestion with an increasing number of participants. However, given the 5 s block time of the private Ethereum network used in this work, and the number of participants for the use case validation scenario, this limitation was not explored in detail. Future research could investigate ways to optimize the gas usage of the smart contract or explore alternative solutions to address scalability and congestion issues.



Block usage by contract operation

Figure 14. Percentage of block usage per smart contract operation.

7. Discussion and Limitations

The implemented energy trading platform successfully demonstrates the feasibility of a permissioned blockchain network for energy trading. The platform was designed to support energy trading between prosumers and consumers, with the support of an MCC. While the use of an access control list in conjunction with oracles allows the energy trading smart contract to access energy information, the relationship is established between the consumer or prosumer and the contract, instead of using IoT devices as a separate entity in the interaction. Recent developments in the field propose the use of Non-Fungible Tokens (NFTs) as a digital representation or "digital twin" of IoT devices. These tokens, standardized by Ethereum under the ERC-721 and ERC-1155 interfaces, enable the establishment of ownership relationships between the householder, consumer, or prosumer and their energy data-collecting devices. This feature includes the data access functionality, which can simplify the addition of further implementations, such as the one presented here.

In terms of cost analysis, the results indicate that a private implementation of the energy trading platform is more cost-effective than implementing the same business logic in the public Ethereum network. This is mainly due to the public network's high transactional fee and longer validation block times, caused by network congestion and a different consensus mechanism. Therefore, it is recommended to implement such platforms in private Ethereum networks or shared private networks, such as consortiums involving multiple entities related to smart grid applications. If by any means there is a requirement to deploy such a platform in a public network, it is recommended to make a prospection into layer two (L2) solutions, which present lower transactional fees and faster validation block time.

The privacy of the energy trading system is achieved through the use of a permissioned blockchain network, which restricts access to only authorized nodes. However, it is important to note that while the access control list and the oracles provide privacy for off-chain data access, the transactions that occur within the energy trading smart contract are still recorded on the blockchain. This includes the funds transferred between parties during the settlement phase and the readings from the off-chain data that are accessed via the oracles. Therefore, further research is needed to ensure the privacy of on-chain transactions in the energy trading system. For example, a technology that could be used to enhance the privacy and security of energy trading platforms is the zero-knowledge proof (Zk-proofs). Zk-proofs enable the verification of data without revealing the data itself. This could be useful in scenarios where the privacy of the energy trading data is of utmost importance, such as trading between competitors or sensitive business transactions.

The analysis of the gas usage of the smart contract revealed that the constructor had an average gas usage of 9%, while the average gas usage for each function call was 0.5%. Although the implementation was successful in supporting the energy trading use case, the gas usage results suggest that there is room for improvement in the smart contract design. Future work could explore the use of more advanced smart contract technologies, such as verifiable computing (VC) or zero-knowledge proof (ZKP), to optimize gas usage and enable the implementation of more complex business logic.

In this work, the old transactions are still kept on the blockchain. However, the concept of offloading can be applied, but a different approach is needed regarding the blockchain network configuration, IoT data, transactions executed by devices, and energy trading participants. For example, in [49], the authors introduced using different types of nodes including full nodes, semi-full nodes, and light nodes, while in [50], the authors developed a hierarchical blockchain storage structure for industrial IoT architectures.

Regarding the law and regulations, in Chile, the current regulation allows prosumers with PV systems to only inject surplus energy to the power grid; however, consumers and prosumers are not allowed to conduct direct energy trading with one another. To enable such opportunities, new regulations should be developed that will bring benefits not only to utility companies but also all participants, including prosumers, consumers, and utilities.

8. Conclusions

This paper presents the development and deployment of a microgrid-based local peerto-peer energy trading system in a laboratory environment. It thoroughly examined and discussed the conceptual architecture of a distributed peer-to-peer network. To establish a blockchain-based system in a test environment, a Chainlink oracle and a private Ethereum blockchain network were employed as foundational components. The research focused on exploring the capabilities of oracles as intermediaries between off-chain data sources, such as smart meter energy readings, and the business logic implemented within smart contracts, which typically lack the ability to directly request off-chain data. Overall, this work highlights the potential of IoT and blockchain technologies to enable sustainable and efficient energy management practices. Despite the promising results, the proposed platform has some limitations and challenges that require further research and development. The following are the main directions to extend the current work:

- To support a large number of participants, there is a need to optimize the platform performance and scalability. This includes optimizing the smart contract code, data storage, and processing algorithms.
- To support real-time energy trading, different business models and trading schemes should be developed where participants trade energy in real time based on the current market conditions.
- Cybersecurity and privacy are important factors to be considered. This includes the data exchange among different participants as well as the sensors, smart meters, and networking infrastructure.

Author Contributions: Conceptualization, M.A.A., J.M.M. and F.C.; methodology, M.A.A., J.M.M., F.C., P.F., A.M.E. and Y.-C.K.; software, F.C.; validation, M.A.A., J.M.M. and F.C.; formal analysis, M.A.A., J.M.M., F.C., P.F., A.M.E. and Y.-C.K.; investigation, M.A.A., J.M.M., F.C., P.F., A.M.E. and Y.-C.K.; writing—original draft preparation, F.C. and M.A.A.; writing—review and editing, M.A.A., J.M.M., F.C., P.F., A.M.E. and Y.-C.K.; funding acquisition, M.A.A. and Y.-C.K. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported in part by the Agencia Nacional de Investigación y Desarrollo (ANID) through the Proyecto Fondecyt de Iniciación en Investigación 2020 under Project ID11200178, DGIIP-UTFSM Chile, and in part by the National Research Foundation of Korea (NSF) Grant through the Korean Government under Grant (2021R1I1A305872911).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors would like to acknowledge Proyecto Fondecyt de Iniciacion en Investigacion 2020 under Project ID11200178, DGIIP-UTFSM Chile, and National Research Foundation of Korea (NSF) Grant through the Korean Government under Grant (2021R1I1A305872911).

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Chainlink Operator Nodes Bridges Transactions Jobs Runs Chains **\$ 6** Definition Сору type = "directrequest" schemaVersion = 1 name = "HEMS API Request externalJobID = "5d9708dd-026d-4410-966d-d871195b30be maxTaskDuration = "0s" contractAddress = "0x59E4fD714b73B733cD8d1c66f82238e087257C29" minIncomingConfirmations = 0 observationSource decode_log [type="ethabidecodelog" abi="OracleRequest(bytes32 indexed specId, address requester, bytes32 requestId, uint256 payment, address callbackAddr, bytes4 callbackFunctionId, uint256 cancelExpiration, uir data="\$(jobRun.logData)" topics="\$(jobRun.logTopics)"] decode_cbor [type="cborparse" data="\$(decode_log.data)"] ds [type="http" method=GET url="{service_url}"] ds_parse ctype="jonparse" path="value"] encode_data [type="thabiencode" abi="(uint256 value)" data="{ \\"value\\": \$(ds_parse) }"] encode_tx [type="ethabiencode" considencies abi="fulfillOracleRequest(bytes32 requestId, uint256 payment, address callbackAddress, bytes4 callbackFunctionId, uint256 expiration, bytes32 data)" data="{\\"requestId\\": \$(decode_log.requestId), \\"payment\\": \$(decode_log.payment), \\"callbackAddress\\": \$(decode_log.callbackAddr), \\"callbackFunctionId\\": \$(decode_log submit_tx [type="ethtx" to="0x59E4fD714b73B733cD8d1c66f82238e087257C29" data="\$(encode_tx)"] decode_log -> decode_cbor -> ds -> ds_parse -> encode_data -> encode_tx -> submit_tx

Figure A1. Job definition in the oracle node operator.



Figure A2. Log message from granting permission operation.



Figure A3. Log message from a new energy trading window initialization.



Figure A4. Error log when prosumer is offering more energy than available.



Figure A5. Error log when the energy trading smart contract does not have permission to request energy data from smart meters.



Figure A6. Log from the results of the matching phase.



Figure A7. Log from the oracle request for obtaining the received energy by the smart meter of the consumer.

	<pre>"from": "0x46cb39166e52d7Ea8077CAa6a764C5507e813FA6", "topic": "0x7bdd298fee8520ef5fb09d30ad01cd090f764b284bf6ee61c6fa184132705f6b", "event": "SettlementState",</pre>
	"0": "0xb45CFB124677b22553e88b2B6189bD1055aC7bE7", "1": "0x46cb39166e52d7Ea8077CAa6a764C5507e813FA6", "2": "432000000000000",
	"prosumer": "0xb45CFB124677b22553e88b2B6189bD1055aC7bE7", "lockedFundsContract": "0x46cb39166e52d7Ea8077CAa6a764C5507e813FA6", "payment": "432000000000000"
	}

Figure A8. Log from the settlement phase, where the prosumer is paid proportionally to what we provided.

References

- 1. Bouhafs, F.; Mackay, M.; Merabti, M. Links to the Future: Communication Requirements and Challenges in the Smart Grid. *IEEE Power Energy Mag.* **2012**, *10*, 24–32. [CrossRef]
- Bayram, I.S.; Shakir, M.Z.; Abdallah, M.; Qaraqe, K. A survey on energy trading in smart grid. In Proceedings of the 2014 IEEE Global Conference on Signal and Information Processing (GlobalSIP), Atlanta, GA, USA, 3–5 December 2014. [CrossRef]
- 3. Hamari, J.; Sjölint, M.; Ukkonen, A. The sharing economy: Why people participate in collaborative consumption. *J. Assoc. Inf. Sci. Technol.* **2015**, *67*, 2047–2059. [CrossRef]
- 4. Jamil, F.; Iqbal, N.; Ahmad, S.; Kim, D. Peer-to-Peer Energy Trading Mechanism Based on Blockchain and Machine Learning for Sustainable Electrical Power Supply in Smart Grid. *IEEE Access* **2021**, *9*, 39193–39217. [CrossRef]
- 5. Kabalci, Y.; Kabalci, E.; Padmanaban, S.; Holm-Nielsen, J.B.; Blaabjerg, F. Internet of Things Applications as Energy Internet in Smart Grids and Smart Environments. *Electronics* **2019**, *8*, 972. [CrossRef]
- 6. Tan, S.; Wu, Y.; Xie, P.; Guerrero, J.M.; Vasquez, J.C.; Abusorrah, A. New Challenges in the Design of Microgrid Systems: Communication Networks, Cyberattacks, and Resilience. *IEEE Electrif. Mag.* **2020**, *8*, 98–106. [CrossRef]
- Wu, Y.; Wu, Y.; Guerrero, J.M.; Vasquez, J.C.; Palacios-Garcia, E.J.; Li, J. Convergence and Interoperability for the Energy Internet: From Ubiquitous Connection to Distributed Automation. *IEEE Ind. Electron. Mag.* 2020, 14, 91–105. [CrossRef]
- 8. Leitao, J.; Gil, P.; Ribeiro, B.; Cardoso, A. A Survey on Home Energy Management. IEEE Access 2020, 8, 5699–5722. [CrossRef]
- Zafar, U.; Bayhan, S.; Sanfilippo, A. Home Energy Management System Concepts, Configurations, and Technologies for the Smart Grid. *IEEE Access* 2020, *8*, 119271–119286. [CrossRef]
- 10. Llaria, A.; Dos Santos, J.; Terrasson, G.; Boussaada, Z.; Merlo, C.; Curea, O. Intelligent Buildings in Smart Grids: A Survey on Security and Privacy Issues Related to Energy Management. *Energies* **2021**, *14*, 2733. [CrossRef]
- 11. Chander, B.; Kumaravelan, G. Internet of Things: Foundation. In *Principles of Internet of Things (IoT) Ecosystem: Insight Paradigm;* Peng, S.L., Pal, S., Huang, L., Eds.; Springer International Publishing: Cham, Switzerland, 2020; pp. 3–33. [CrossRef]
- 12. Zhang, C.; Wu, J.; Cheng, M.; Zhou, Y.; Long, C. A Bidding System for Peer-to-Peer Energy Trading in a Grid-connected Microgrid. *Energy Procedia* **2016**, 103, 147–152. [CrossRef]
- 13. Mengelkamp, E.; Gärttner, J.; Rock, K.; Kessler, S.; Orsini, L.; Weinhardt, C. Designing microgrid energy markets: A case study: The Brooklyn Microgrid. *Appl. Energy* **2018**, *210*, 870–880. [CrossRef]
- 14. Zhou, Y.; Wu, J.; Long, C.; Ming, W. State-of-the-Art Analysis and Perspectives for Peer-to-Peer Energy Trading. *Engineering* **2020**, *6*, 739–753. [CrossRef]
- 15. Soto, E.A.; Bosman, L.B.; Wollega, E.; Leon-Salas, W.D. Peer-to-peer energy trading: A review of the literature. *Appl. Energy* **2021**, 283, 116268. [CrossRef]
- 16. Wu, Y.; Wu, Y.; Guerrero, J.M.; Vasquez, J.C. Digitalization and decentralization driving transactive energy Internet: Key technologies and infrastructures. *Int. J. Electr. Power Energy Syst.* **2021**, *126*, 106593. [CrossRef]
- Condon, F.; Martínez, J.M.; Kim, Y.C.; Ahmed, M.A. EnergyAuction: oracle Blockchain-Based Energy Trading System for Microgrids. In Proceedings of the 2023 IEEE Conference on Power Electronics and Renewable Energy (CPERE), Luxor, Egypt, 15–17 February 2023; pp. 1–6.
- 18. Mengelkamp, E.; Notheisen, B.; Beer, C.; Dauer, D.; Weinhardt, C. A blockchain-based smart grid: Towards sustainable local energy markets. *Comput. Sci. Res. Dev.* 2018, 33, 207–214. [CrossRef]
- Luo, Y.; Itaya, S.; Nakamura, S.; Davis, P. Autonomous cooperative energy trading between prosumers for microgrid systems. In Proceedings of the 39th Annual IEEE Conference on Local Computer Networks Workshops, Edmonton, AB, Canada, 8–11 September 2014. [CrossRef]

- 20. Zhang, C.; Wu, J.; Zhou, Y.; Cheng, M.; Long, C. Peer-to-Peer energy trading in a Microgrid. *Appl. Energy* 2018, 220, 1–12. . [CrossRef]
- Andoni, M.; Robu, V.; Flynn, D.; Abram, S.; Geach, D.; Jenkins, D.; McCallum, P.; Peacock, A. Blockchain technology in the energy sector: A systematic review of challenges and opportunities. *Renew. Sustain. Energy Rev.* 2019, 100, 143–174. [CrossRef]
- Li, Z.; Kang, J.; Yu, R.; Ye, D.; Deng, Q.; Zhang, Y. Consortium Blockchain for Secure Energy Trading in Industrial Internet of Things. *IEEE Trans. Ind. Inform.* 2018, 14, 3690–3700. [CrossRef]
- 23. Aitzhan, N.Z.; Svetinovic, D. Security and Privacy in Decentralized Energy Trading Through Multi-Signatures, Blockchain and Anonymous Messaging Streams. *IEEE Trans. Dependable Secur. Comput.* **2018**, *15*, 840–852. [CrossRef]
- Wang, S.; Sheng, H.; Zhang, Y.; Yang, D.; Shen, J.; Chen, R. Blockchain-Empowered Distributed Multi-Camera Multi-Target Tracking in Edge Computing. *IEEE Trans. Ind. Inform.* 2023, 1–10. [CrossRef]
- Cao, B.; Wang, X.; Zhang, W.; Song, H.; Lv, Z. A Many-Objective Optimization Model of Industrial Internet of Things Based on Private Blockchain. *IEEE Netw.* 2020, 34, 78–83. [CrossRef]
- Mihaylov, M.; Jurado, S.; Avellana, N.; Van Moffaert, K.; de Abril, I.M.; Nowé, A. NRGcoin: Virtual currency for trading of renewable energy in smart grids. In Proceedings of the 11th International Conference on the European Energy Market (EEM14), Krakow, Poland, 28–30 May 2014; pp. 1–6. [CrossRef]
- 27. Wang, W.; Hoang, D.T.; Hu, P.; Xiong, Z.; Niyato, D.; Wang, P.; Wen, Y.; Kim, D.I. A Survey on Consensus Mechanisms and Mining Strategy Management in Blockchain Networks. *IEEE Access* 2019, *7*, 22328–22370. [CrossRef]
- Liu, C.; Zhang, X.; Chai, K.K.; Loo, J.; Chen, Y. A survey on blockchain-enabled smart grids: Advances, applications and challenges. *IET Smart Cities* 2021, 3, 56–78.
- 29. Szabo, N. Formalizing and securing relationships on public networks. First Monday 1997, 2. [CrossRef]
- 30. Caldarelli, G. Overview of Blockchain oracle Research. Future Internet 2022, 14, 175. [CrossRef]
- 31. Xu, X.; Pautasso, C.; Zhu, L.; Lu, Q.; Weber, I. A Pattern Collection for Blockchain-Based Applications. In Proceedings of the 23rd European Conference on Pattern Languages of Programs, New York, NY, USA, 19 February 2018.
- Antal, C.; Cioara, T.; Antal, M.; Mihailescu, V.; Mitrea, D.; Anghel, I.; Salomie, I.; Raveduto, G.; Bertoncini, M.; Croce, V.; et al. Blockchain based decentralized local energy flexibility market. *Energy Rep.* 2021, 7, 5269–5288. [CrossRef]
- Al Breiki, H.; Al Qassem, L.; Salah, K.; Habib Ur Rehman, M.; Sevtinovic, D. Decentralized Access Control for IoT Data Using Blockchain and Trusted oracles. In Proceedings of the 2019 IEEE International Conference on Industrial Internet (ICII), Orlando, FL, USA, 11–12 November 2019; pp. 248–257. [CrossRef]
- Wu, Y.; Wu, Y.; Cimen, H.; Vasquez, J.C.; Guerrero, J.M. Towards collective energy Community: Potential roles of microgrid and blockchain to go beyond P2P energy trading. *Appl. Energy* 2022, 314, 119003. [CrossRef]
- 35. Pradhan, N.R.; Singh, A.P.; Verma, S.; Kavita; Wozniak, M.; Shafi, J.; Ijaz, M.F. A blockchain based lightweight peer-to-peer energy trading framework for secured high throughput micro-transactions. *Sci. Rep.* **2022**, *12*, 14523. [CrossRef]
- Appasani, B.; Mishra, S.K.; Jha, A.V.; Mishra, S.K.; Enescu, F.M.; Sorlei, I.S.; Bîrleanu, F.G.; Takorabet, N.; Thounthong, P.; Bizon, N. Blockchain-Enabled Smart Grid Applications: Architecture, Challenges, and Solutions. *Sustainability* 2022, 14, 8801. [CrossRef]
- Capper, T.; Gorbatcheva, A.; Mustafa, M.A.; Bahloul, M.; Schwidtal, J.M.; Chitchyan, R.; Andoni, M.; Robu, V.; Montakhabi, M.; Scott, I.J.; et al. Peer-to-peer, community self-consumption, and transactive energy: A systematic literature review of local energy market models. *Renew. Sustain. Energy Rev.* 2022, 162, 112403. [CrossRef]
- 38. Frieden, D.; Tuerk, A.; Neumann, C.; d'Herbemont, S.; Roberts, J. Collective self-consumption and energy communities: Trends and challenges in the transposition of the EU framework. *Compil. Integr. Community Power Energy Isl.* **2020**. [CrossRef]
- Chen, S.; Liu, C.C. From demand response to transactive energy: State of the art. J. Mod. Power Syst. Clean Energy 2017, 5, 10–19. [CrossRef]
- 40. Parag, Y.; Sovacool, B.K. Electricity market design for the prosumer era. Nat. Energy 2016, 1, 16032. [CrossRef]
- 41. Wang, Q.; Zhang, C.; Ding, Y.; Xydis, G.; Wang, J.; Østergaard, J. Review of real-time electricity markets for integrating Distributed Energy Resources and Demand Response. *Appl. Energy* **2015**, *138*, 695–706. [CrossRef]
- Koirala, B.P.; Koliou, E.; Friege, J.; Hakvoort, R.A.; Herder, P.M. Energetic communities for community energy: A review of key issues and trends shaping integrated community energy systems. *Renew. Sustain. Energy Rev.* 2016, 56, 722–744. [CrossRef]
- 43. Saad, W.; Han, Z.; Poor, H.V.; Basar, T. Game-Theoretic Methods for the Smart Grid: An Overview of Microgrid Systems, Demand-Side Management, and Smart Grid Communications. *IEEE Signal Process. Mag.* **2012**, *29*, 86–105. [CrossRef]
- 44. Abdella, J.; Shuaib, K. Peer to Peer Distributed Energy Trading in Smart Grids: A Survey. Energies 2018, 11, 1560. [CrossRef]
- 45. Mohanta, B.K.; Panda, S.S.; Jena, D. An Overview of Smart Contract and Use Cases in Blockchain Technology. In Proceedings of the 2018 9th International Conference on Computing, Communication and Networking Technologies (ICCCNT), Bengaluru, India, 10–12 July 2018; pp. 1–4. [CrossRef]
- Al-Breiki, H.; Rehman, M.H.U.; Salah, K.; Svetinovic, D. Trustworthy Blockchain oracles: Review, Comparison, and Open Research Challenges. *IEEE Access* 2020, *8*, 85675–85685. [CrossRef]
- 47. Saltini, R.; Hyland-Wood, D. Ibft 2.0: A safe and live variation of the ibft blockchain consensus protocol for eventually synchronous networks. *arXiv* **2019**, arXiv:1909.10194.
- Fan, C.; Lin, C.; Khazaei, H.; Musilek, P. Performance Analysis of Hyperledger Besu in Private Blockchain. In Proceedings of the 2022 IEEE International Conference on Decentralized Applications and Infrastructures (DAPPS), San Francisco, CA, USA, 15–18 August 2022; pp. 64–73. [CrossRef]

- 49. Zhao, P.; Cheng, H.; Fang, Y.; Wang, X. A Secure Storage Strategy for Blockchain Based on MCMC Algorithm. *IEEE Access* 2020, *8*, 160815–160824. [CrossRef]
- Wang, G.; Shi, Z.; Nixon, M.; Han, S. ChainSplitter: Towards Blockchain-Based Industrial IoT Architecture for Supporting Hierarchical Storage. In Proceedings of the 2019 IEEE International Conference on Blockchain (Blockchain), Atlanta, GA, USA, 14–17 July 2019; pp. 166–175. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.